Author's response to reviews

Title: The feasibility and effectiveness of high-intensity boxing training versus moderate-intensity brisk walking in adults with abdominal obesity: A pilot study

Authors:
Birinder S Cheema (B.Cheema@uws.edu.au)
Timothy B Davies (t.davies@hotmail.com.au)
Matthew Stewart (matthew.stewart@student.bond.edu.au)
Shona Papalia (S.Papalia@uws.edu.au)
Evan Atlantis (E.Atlantis@uws.edu.au)

Version: 2
Date: 28 November 2014

Author's response to reviews:

Please see attached letter.